Author’s response to reviews

Title: Filtering Data from the Collaborative Initial Glaucoma Treatment Study for Improved Identification of Glaucoma Progression

Authors:

Greggory J Schell (schellg@umich.edu)
Mariel S Lavieri (lavieri@umich.edu)
Joshua D Stein (jdstein@umich.edu)
David C Musch (dmusch@umich.edu)

Version: 2 Date: 2 March 2013

Author’s response to reviews:

The revisions made were to address the requested ethics statement. We included the IRB approval.